EAS 305 Quiz 2, Spring 1998 Page 1 of 1

Answer the following problems. (5 points eac h), 55 minutes.

1- The life-time (in hours) of a device is a random variable with probability density: , . Find the probability that the device will still be in operation after hours, given that it survived the first hour, where .

2- A manufacturer produces memory chips using two different processes with time-to-failure with pdf of , and , respectively. Assume independence.

Determine the probability the time-to-failure is greater for chips characterized by than it is form chips characterized by .

fT t( ) α e α t– u t( ) = α 0> T_2 T_1 T_2 T_1>

fX x( ) ae ax– u x( ) a, 0>

= fY y( ) be by– u y( ) b 0>, = X Y